Author's response to reviews

Title: Invasive meningococcal disease epidemiology and control measures: a framework for evaluation

Authors:

Jamie J Caro (jcaro@caroresearch.com)
Jorgen Moller (jmoller@caroresearch.com)
Denis Getsios (dgetsios@caroresearch.com)
Laurent Coudeville (Laurent.Coudeville@sanofipasteur.com)
Wissam El-Hadi (wissam.el-hadi@sanofi-aventis.com)
Catherine Chevat (catherine.chevat@sanofi-pasteur.com)
Van Hung Nguyen (vanhung.nguyen@aventis.com)
Ingrid Caro (icaro@caroresearch.com)

Version: 4 Date: 22 June 2007

Author's response to reviews: see over
June 22, 2007

The BioMed Central Editorial Production Team
BMC-series Journals
BioMed Central
Middlesex House
34-42 Cleveland Street
London W1T 4LB

Dear Sirs:

Please find enclosed the revised version of our manuscript entitled “Invasive meningococcal disease epidemiology and control measures: a framework for evaluation” (MS: 5513254931104146).

All of the formatting changes requested have been addressed as per instructions in your e-mail dated June 20, 2007.

Thank you for your consideration of this manuscript.

Sincerely,

[Signature]

J. Jaime Caro, MDCM, FRCPC, FACP